Author’s response to reviews

Title: Self-Reported Psychopathy in the Middle East: A Cross-National Comparison across Egypt, Saudi Arabia, and the United States

Authors:
 Robert Latzman (rlatzman@gsu.edu)
 Ahmed Megreya (amegreya@qu.edu.qa)
 Lisa Hecht (lhecht1@student.gsu.edu)
 Joshua Miller (jdmiller@uga.edu)
 Anne Winiarski (anne.winiarski@emory.edu)
 Scott Lilienfeld (slilien@emory.edu)

Version: 3 Date: 20 Oct 2015

Author’s response to reviews:

RESPONSES TO EDITOR’S REQUESTS:

1. The full name of the ethics committee in the USA has been provided (see pg. 9).

2. The details of the ethics approval in Saudi Arabia and Egypt have been provided (see pg. 9).

3. It is now clearly stated in the Methods section that written informed consent was obtained from all participants across the three countries (see pg. 9).

4. The Title Page at the front of the manuscript now includes names, institutions, countries, and email addresses of all authors as well as the full postal address of the submitting and corresponding author.